Student project { Lecture Notes

ffl Lecture notes which reflect your own ideas and which clarify your questions and problems with the material can only be written by yourselves.

ffl Organization in groups reduces the individual work load and increases communication and contacts in your class as well as with the lecturer.

ffl Based on personal notes, text books, and notes of the lecturer you will write your own script and, almost without realizing it, you will thoroughly revise, learn and understand the lecture material.

Tasks:

ffl Reformulation of notes, additional comments and examples ffl Design of useful figures supporting the explanations ffl Text processing, setting up an index ffl Organization of groups and distribution of results via Internet ffl Proof reading ffl Regular editorial meetings

Advantages

ffl You will get used to writing scientific texts ffl Thorough revision of the lecture material is a side effect ffl Publication as AG Technomathematik report ffl Final lecture notes will suit your taste!

First meeting: